Author's response to reviews

Title: Hospital stay in patients admitted for acute bipolar manic episodes prescribed quetiapine immediate or extended release: a retrospective non-interventional cohort study (HOME)

Authors:

Oguz Karamustafalioglu (oguzkaramustafa@superonline.com)
Andreas Reif (Reif_A@klinik.uni-wuerzburg.de)
Murad Atmaca (matmaca_p@yahoo.com)
Domingo Gonzalez (Domingo.gonzalez@bsmhft.nhs.uk)
Miriam Moreno-Manzanaro (Miriam.MORENO-MANZANARO@astrazeneca.com)
Miguel Angel Gonzalez (miguelangel.gonzalez@quintiles.com)
Esteban Medina (esteb.medina@gmail.com)
Antonello Bellomo (a.bellomo@unifg.it)

Version: 2  Date: 13 January 2014

Author's response to reviews:

The ethics committee information has been added to the manuscript, as requested.